ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Vey, Meike; Namockel, Nils; Ruhnau, Oliver

Working Paper Green charged decision-making: How two-part remuneration, contract flexibility, and environmental nudging drive vehicle-to-grid participation

EWI Working Paper, No. 01/25

Provided in Cooperation with: Institute of Energy Economics at the University of Cologne (EWI)

Suggested Citation: Vey, Meike; Namockel, Nils; Ruhnau, Oliver (2025) : Green charged decisionmaking: How two-part remuneration, contract flexibility, and environmental nudging drive vehicleto-grid participation, EWI Working Paper, No. 01/25, Institute of Energy Economics at the University of Cologne (EWI), Cologne

This Version is available at: https://hdl.handle.net/10419/315027

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Green charged decision-making: How two-part remuneration, contract flexibility, and environmental nudging drive vehicle-to-grid participation

AUTHORS

Meike Vey Nils Namockel Oliver Ruhnau

EWI Working Paper, No 01/25

January 2025

Institute of Energy Economics at the University of Cologne (EWI) www.ewi.uni-koeln.de

Institute of Energy Economics at the University of Cologne (EWI)

Alte Wagenfabrik Vogelsanger Str. 321a 50827 Köln Germany

Tel.: +49 (0)221 277 29-100 Fax: +49 (0)221 277 29-400 www.ewi.uni-koeln.de

CORRESPONDING AUTHOR

Meike Vey meike.vey@web.de

ISSN: 1862-3808

The responsibility for working papers lies solely with the authors. Any views expressed are those of the authors and do not necessarily represent those of the EWI.

Green charged decision-making: How two-part remuneration, contract flexibility, and environmental nudging drive vehicle-to-grid participation

Meike Vey^{a,*}, Nils Namockel^{a,b}, Oliver Ruhnau^{a,b}

^aInstitute of Energy Economics at the University of Cologne, Vogelsanger Strasse 321a, 50827 Cologne, Germany. ^bChair of Energy Economics, University of Cologne, Vogelsanger Strasse 321a, 50827 Cologne, Germany.

Abstract

Electric mobility and renewable energy play key roles in the global energy transition. In this context, vehicleto-grid technology, which enables bidirectional energy flow between electric vehicles and the grid, could make electric vehicles usable as energy storage units, thus supporting grid stability and integration of renewable energy. However, the willingness of electric vehicle owners to participate in vehicle-to-grid contracts remains insufficiently understood, particularly regarding how they respond to specific contract attributes. This paper addresses this gap by conducting a discrete choice experiment to evaluate the preferences of current and potential future German electric vehicle drivers for various vehicle-to-grid contract alternatives. We find that cycle-based remuneration, flexible contract duration, and environmental nudging significantly enhance consumer acceptance. Conversely, a lower guaranteed battery level and longer minimum plug-in durations negatively impact participation. We also test how respondent characteristics influence participation and identify income-dependent preferences, such as lower-income individuals attributing a stronger preference to fixed daily payments than higher-income individuals. Our differentiated findings may be used to improve contract designs and marketing efforts to address the unique V2G preferences of various user segments. *Keywords:* Vehicle-to-grid, Discrete choice experiment, Willingness to accept, Preferences, Bidirectional charging

JEL classification: C25, D12, Q42, Q48, Q51, Q58, R41

 $[\]label{eq:corresponding} \mbox{``erresponding author, $email: meike.vey@web.de.''}$

1. Introduction

The rapid adoption of electric vehicles (EVs) will closely link the transportation sector to the electricity sector. For the example of Europe, EVs are projected to reach 60% of car sales by 2030, following the *Stated Policies* scenario [1]. On the one hand, this development may help decarbonize the transportation sector by using electricity from renewable energy sources (RES). On the other hand, flexible charging and discharging of EVs may help address challenges associated with the intermittency of the most prominent RES, namely wind and solar energy. In particular, vehicle-to-grid (V2G) technology¹, which enables bidirectional energy flow between EVs and the grid, holds the potential to leverage EVs as mobile energy storage, capable of absorbing surplus RES and discharging it back into the grid when RES are scarce, thereby reducing reliance on conventional power plants [3, 2].

The theoretical potential of V2G technology is substantial. With EVs parked approximately 97% of the time, their collective storage capacity offers a promising alternative to stationary storage systems [4, 5]. V2G technology provides several benefits, including economic savings on electricity bills and system costs, reductions in CO_2 emissions, and improved grid reliability [6, 7, 8]. For instance, a recent study [7] estimates that V2G can reduce the average price paid for EV charging by 28–67% versus unidirectional charging with a time-invariant tariff. While V2G technology comes at additional costs, car manufacturers are increasingly developing V2G-compatible vehicles, suggesting that benefits are expected to outweigh these costs [e.g. 9, 10, 11, 12].

Despite this theoretical potential, regulatory, technical, and socioeconomic barriers currently hinder the commercial viability of V2G. The utilization of V2G technology requires smart meters, which are still rare in some countries [13]. Regulation for feeding electricity into the grid, such as grid codes, has been developed for large power producers and may be unsuited for distributed V2G systems. Electricity taxes may make it uneconomical to sell energy back to the grid [14]. Finally, even when these issues are solved, V2G-capable charging infrastructure may be lacking [15].

While much of the existing research on V2G focuses on techno-economic aspects, consumer acceptance has remained underexplored. Between 2015 and 2017, only 3% of peer-reviewed studies on V2G addressed consumer acceptance [16]. Similarly, Park Lee [17] observed that socio-technical barriers receive insufficient attention in the literature, noting a lack of knowledge about the relationship between contracts, system operations, and actor behavior. Furthermore, a report by Everoze EVConsult [18] highlights that, while

 $^{^{1}}$ V2G is part of a broader suite of 'vehicle-to-x' technologies, including vehicle-to-home, where EVs supply power to homes, and vehicle-to-vehicle, where energy is shared between EVs [2].

98% of V2G projects focused on technical factors, only 27% considered social factors, leaving a significant gap in understanding consumer behavior.

This article aims to enhance the limited understanding of consumer preferences for V2G contracts with a discrete choice experiment (DCE) in the context of Germany. Within the DCE, we focus on the effects of flexible contract design, a remuneration scheme that combines fixed and variable payments, and environmental nudging on the willingness to participate in V2G. Furthermore, we differentiate consumer preferences by income. The results will inform economic modeling and practical applications, namely the effective design of contracts between EV owners and aggregators.

Our research contributes to two strands of the existing V2G literature: contract design and consumer acceptance. Within the first strand, some studies apply contract and game theory to determine the optimal design of contracts between aggregators and EV owners in V2G systems. For example, Gao et al. [19] demonstrate that aggregators can determine an optimal unit price for EV charging to maximize profits under asymmetric information. Similarly, Jember et al. [20] implement a non-cooperative Stackelberg game to identify an optimal pricing and discharging strategy, and Zhang et al. [21] model a system in which EVs can submit bids for charging and discharging to a profit-maximizing broker. Other studies derive contract designs from real-world applications. For instance, Guille and Gross [22] propose a framework in which aggregators manage EV batteries as controllable loads, offering charging discounts in exchange for scheduled grid connections. Jiao et al. [23] focus on revenue-sharing contracts between the aggregator, EV owner, and grid, aiming to maximize collective profit.

The second strand of literature evaluates consumer preferences in V2G contexts. A pioneering study by Parsons et al. [24] investigated different V2G remuneration models in combination with guaranteed minimum driving ranges and plug-in durations, finding that remuneration was the most influential attribute, with flexible performance-based payments or upfront discounts emerging as the most effective strategies to encourage participation. By contrast, more recent studies [25, 26, 27] find that concerns about vehicle availability and battery degradation often outweigh financial incentives when deciding on V2G participation. Huang et al. [27] identify fast-charging options as an effective countermeasure to availability concerns and show that V2G contracts with a pre-defined green electricity mix benefited from increased consumer acceptance.

We aim to make four significant contributions to this existing literature. First, we introduce an innovative remuneration structure, which combines a fixed per-day with a variable per-kWh component, and analyze the acceptance of these components depending on the participants' income. This approach may address battery degradation concerns as additional revenue is generated whenever the battery is used. Second, we propose V2G contracts that can be adjusted daily and compare their acceptance to monthly adjustable agreements. We argue that such contract flexibility could address availability concerns while providing certainty for aggregators relative to fast-charging options. Third, we update the findings of previous studies. In particular, we find that minimum battery levels may not be as important as previously thought, which we explain by technological progress in EV battery technology. Finally, we assess the influence of environmental nudging with an A/B test. Such nudging could leverage consumers' growing concern for sustainability to benefit V2G, similar to the positive effects found for other types of pro-environmental behavior [28, 29, e.g.].

This paper is structured as follows: Section 2 outlines the design and implementation of the DCE. Section 3 details the approach to estimating the willingness to participate in V2G contracts. Section 4 presents the findings, including statistical analysis and economic implications. Section 5 contextualizes the results within the existing literature. Section 6 summarizes the results, derives recommendations, and discusses further research.

2. Conceptualization of the DCE

This section outlines the design and implementation of the DCE. It starts with Subsection 2.1 defining the V2G contract attributes investigated in the DCE and continues with Subsection 2.2 describing the design and implementation of the DCE, including the nudging techniques used.

2.1. Definition of V2G contract attributes

Identifying the relevant attributes and their levels is crucial for designing an effective DCE. The challenge is representing all essential aspects of the decision while keeping the number of attributes and levels manageable to avoid overburdening participants. The following discusses how we derived our selection of attributes and levels based on recent literature. Additionally, we took practical considerations into account to ensure that the chosen levels are relevant in real-world conditions. Table 1 provides an overview of our selected attributes and levels.

Attribute	Levels
Fixed remuneration	0 cents/day, 25 cents/day, 50 cents/day
Variable remuneration	5 cents/kWh, 10 cents/kWh, 15 cents/kWh
Contract flexibility	Daily, Monthly
Minimum plug-in duration	5 hours, 10 hours, 15 hours
Minimum battery level	35%, 50%, 70%

Table 1: Summary of contract attributes and their levels

Remuneration: Remuneration is critical in determining whether EV owners will engage in V2G. In this study, two types of remuneration are included: a fixed payment, which rewards participants for making their vehicle available to the grid at rates of 0, 25, and 50 cents per day, and a variable payment, which rewards participants at rates of 5, 10, and 15 cents per kWh discharged to the grid. The fixed payment structure is based on capacity market models and is supported by previous literature on fixed monthly remuneration in similar markets [25]. The levels of the variable payments are derived from market benchmarks in peak shaving, similar to payments for photovoltaic operators [30, 31]. Higher remuneration in both schemes is expected to increase the willingness to participate.

Contract flexibility: Contract flexibility determines how easily EV owners can adapt their V2G contract to changing circumstances. We test two levels of contract flexibility: daily flexibility, meaning that the minimum plug-in duration and the minimum battery level can be adjusted daily or that the contract can be canceled or paused after any day, and monthly flexibility, meaning that these attributes are fixed for an entire month. This is much less than previous studies proposing contract durations of 6 to 48 months [26, 27]. We argue that such long durations may be impractical for many consumers and expect more flexible contracts (i.e., daily adjustability) to increase participation.

Minimum plug-in duration: The minimum plug-in duration, i.e., the minimum time an EV is connected to the grid, directly affects the ability of V2G systems to manage energy supply. Literature suggests that cars are parked for extended periods [32], which could make it feasible to require a minimum connection time. In this experiment, three levels of such a minimum plug-in duration are tested: 5, 10, and 15 hours per day. While longer durations increase the utility for aggregators, they are expected to reduce EV owners' willingness to participate in V2G due to the inconvenience of reduced vehicle availability.

Minimum battery level: The minimum battery level ensures EV owners retain sufficient charge for emergencies. For this study, three levels are chosen: 35%, 50%, and 70%. These levels account for emergencies on the one hand and aim to balance battery preservation with V2G participation on the other hand. Studies by the German Automobile Association suggest that a 25-kilometer round trip is sufficient to reach essential services like pharmacies [33]. This translates into a minimum battery level of approximately 12.5% in an EV with a 60 kWh capacity, based on an average consumption of 15 kWh per 100 km [34]. Moreover, battery health guidelines recommend maintaining a charge between 20% and 80% [35], which we assume should not be violated even after an emergency drive. Hence, we set the lowest minimum battery level to 35% (12.5% + 20% + 2.5% additional buffer). It is hypothesized that higher battery levels will increase the willingness to participate in V2G, allowing greater flexibility for vehicle use.

2.2. Design and implementation of the DCE

This subsection details the design and implementation of the DCE. It describes the way choice sets were determined, the environmental nudges incorporated to influence participant decision-making, and additional details regarding the implementation of the experiment.

2.2.1. Composition of choice sets

The design of the DCE requires balancing the statistical needs for reliable data analysis with the participant's cognitive load. Presenting too many choice sets can lead to respondent fatigue, while too few sets might fail to capture significant patterns. Presenting all possible combinations of attributes, i.e., a full factorial design, is deemed impractical due to the large number of attributes and levels, which would overwhelm participants [36]. Instead, a D-efficient design was chosen, which maximizes the statistical information from the choice sets [36, 37]. This method ensures a balance between comprehensiveness and respondent burden.

The software Ngene was used to generate the choice sets. Each choice set offers two V2G contract options and a no-choice option if participants do not wish to participate in either option. The design and the attributes with its levels are presented in Appendix A. The composition of the choice sets is based on utility functions, which were informed by general trends in the literature. Additionally, Ngene improves efficiency by using already known parameter values as starting values for the utility functions. These parameters could be derived from previous studies or pilot surveys [37]. For this study, we only integrated the basic tendency (whether participants value the change of an attribute as positive or negative). Otherwise, the differentiation between two different survey versions would lead to two different designs and could, therefore, lead to biased results. Several utility functions were tested, indicating that 15 choice sets allow for high statistical efficiency without overloading participants [38].

2.2.2. Environmental nudging

To examine the effect of environmental nudging, the survey was divided into two versions: A reference survey (ECON) and a survey containing the nudging measures (ENV). A single link was created to randomly redirect participants to the ECON and ENV survey with equal probability.

Both surveys comprised the same choice sets, but in the ENV survey, all V2G contract options were visually highlighted in green to frame them as environmentally positive. Examples for the choice sets in the ECON and the ENV surveys are presented in Appendix B. Furthermore, the participants were shown an introductory video explaining the concept of V2G, which differed by survey type. The video in the ECON survey focused on economic benefits, while the video in the ENV survey emphasized environmental advantages. Both videos have been created with ProCreate and Mango Whiteboard Animator and were roughly two minutes long.² For participants who could not listen to the video, the content was provided in written form. The hypothesis is that this pro-environmental nudging increases participants' willingness to engage in V2G programs, although we cannot distinguish the individual effects of the video or color.³

2.2.3. Further implementation details

Following the video and before answering the choice sets, participants were asked to assume they owned an EV with a 60 kWh battery, had access to a home charging station, and made decisions based on a typical workday. In addition to exploring V2G preferences through choice sets, the survey collected sociodemographic data and insights into participants' attitudes toward technology, EV ownership, and sustainability (see Appendix C).

The survey's target group includes all German individuals over 18 with a driver's license. The choice of distribution channels was optimized to primarily reach EV owners, EV drivers, and people interested in EVs. This choice was motivated by the assumption that these respondents can empathize with the choice situation in the V2G context. The survey was advertised in several EV internet forums and distributed via survey exchange groups and platforms, mailing lists, and personal networks. Furthermore, personal acquaintances were asked to participate and forward the survey to potentially interested individuals.

 $[\]label{eq:link} \ensuremath{^2\text{Link}}\xspace{1.5}\x$

 $^{^{3}}$ In this study we use the term nudging to refer to our implemented non-coercive approaches to influence human behavior, as used in [29]. Alternatively, our implemented measures may also be referred to as framing.

3. Approach to estimate the willingness to participate

This section describes the two models used to derive preferences from stated choices and the underlying utility function.

3.1. Deriving preferences from stated choices

This study estimates preferences for different V2G contracts using choice models grounded in McFadden's Random Utility Theory [39]. This theory provides a robust theoretical framework for analyzing individual decision-making by assuming that individuals choose the option that maximizes their utility. The utility of a product or service is determined by its attributes, and the probability of selecting a specific option depends on its utility relative to other alternatives in the choice set.

We chose the Multinomial Logit (MNL) model and the Mixed Logit (MXL) model to estimate choice probabilities because of their complementary strengths in capturing consumer preferences and addressing the complexities of choice data. The MNL model, detailed in [40], is widely used for its simplicity, computational efficiency, and ease of interpretation. However, it assumes independence of irrelevant alternatives and identical preferences across individuals, which can limit its applicability. The MXL model, described extensively in [41, 40], addresses these limitations. It incorporates random taste variations and allows for correlation in the error term, enabling the capture of heterogeneity in individual preferences. By modeling parameters as random coefficients that follow a specified distribution, the MXL model provides a more flexible and realistic representation of decision-making behavior. MXL models are, however, significantly more complex. More precisely, overfitting and insufficient sample sizes can lead to unreliable results, and convergence issues may arise during estimation. To address these issues, we use the results of the MNL model as a benchmark to validate the robustness of the MXL model and as starting values for the MXL estimation process. Louviere [42] provides detailed derivations, assumptions, and examples that underpin the methodologies applied in this study.

In sum, using both MNL and MXL models allows for a comparative analysis that enhances the robustness and reliability of the results. While the MNL model provides a baseline for understanding general trends, the MXL model offers deeper insights into preference heterogeneity. This combination balances simplicity and the need to account for more complex decision-making patterns.

3.2. Definition of the utility function

The DCE was evaluated using the Biogeme software package in Python. Biogeme implements various analyses on DCEs, particularly using MNL and MXL models as considered in this study [43, 44]. More precisely, we employed the Newton algorithm with a trust region for simple bound constraints to estimate the MXL model [45].

To determine the impact of nudging, the data from both studies is merged, and a dummy variable is introduced. This variable is set to one if the participant completed the ENV survey. Otherwise, it was set to zero.

Subsequently, it was determined which socio-demographic factors to include in the model. To address potential collinearities, a variance inflation factor (VIF) test was conducted, where elevated VIF values indicate dependencies between variables. However, the VIF values for the variables in this study are significantly smaller than the commonly accepted threshold of ten [46]. Thus, sufficient independence between the parameters is assumed, allowing all characteristics to be retained in the analysis.

The response options were grouped into broader categories in some cases to enable statistical analysis, as otherwise there would be too few participants corresponding to each category. The relevant groupings are summarized in Appendix D.

As criteria for the evaluation of the goodness of fit of a model, the Akaike Information Criterion (AIC) and Bayesian Information Criterion (BIC) [47] are used. Furthermore, for the MNL model, we evaluate the McFadden's \mathbb{R}^2 , which ranges from zero to one. A higher value indicates greater predictive power of the model, while typically, a McFadden's \mathbb{R}^2 value between 0.2 and 0.4 is considered indicative of a good fit [39]. Comparing the AIC, BIC, and \mathbb{R}^2 metrics, it is found that almost all parameters are best represented when using dummy variables, except the parameter for fixed remuneration, which can be effectively incorporated as a continuous linear term. The other parameters do not develop linearly, and, given only three data points, performing a meaningful function approximation is not feasible [48]. Possible explanations for this non-linearity are explored in Section 4. The reference categories are detailed in Table 2. For the contract attributes, the most attractive levels were selected as references.

Category	Reference level
Variable remuneration	15 cents/kWh
Contract flexibility	Daily
Minimum plug-in duration	5 hours
Minimum battery level	70%
Age	55 years or older
Gender	Male
Level of education	School degree
Housing situation	Rent
Income	€3,000 or above
Comparing electricity tariffs	Compare
Technical interest	Not interested
Concerned about climate change	Not concerned
Car relationship	No corresponding relationship
Car usage	Frequently
Distance travelled	No distance
EV ownership	Interested
V2G familiarity	Not familiar
V2G concerns	No corresponding concern
Environmental nudging	No (ECON-survey and/or did not watch the video)
Attribute description	Read

Table 2: Reference levels

In addition to the main effects, we tested for significant interaction effects. Only the interaction between remuneration schemes and income turned out to be significant and was, therefore, kept in the utility function. The final utility function with an alternative-specific constant (ASC) is presented in equation (1) for the example of one V2G contract option and the MNL model. The contract attributes from the choice sets and the interaction effects are framed in red and blue, respectively. The utility function for the MXL model excludes the interaction effect due to convergence issues, a frequent limitation of MXL models. The utility functions of all models and alternatives, including the option opt-out, can be found in Appendix E.

$V1 = ASC_{V1}$

 $+\beta_{\text{FixRem}} \cdot \text{FixRem}$ $+\beta_{\text{VarRem5c}} \cdot \text{VarRem}_{5c} + \beta_{\text{VarRem10c}} \cdot \text{VarRem}_{10c}$ $+\beta_{\text{ChangeContractMonthly}} \cdot \text{ChangeContractMonthly}$ $+\beta_{\mathrm{MinPlugIn10h}} \cdot \mathrm{MinPlugIn}_{10h} + \beta_{\mathrm{MinPlugIn15h}} \cdot \mathrm{MinPlugIn}_{15h}$ $+\beta_{\text{MinBattery35p}} \cdot \text{MinBatteryLevel}_{35p} + \beta_{\text{MinBattery50p}} \cdot \text{MinBatteryLevel}_{50p}$ $+ \beta_{\mathrm{Age18-34}} \cdot \mathrm{Age}_{18-34} + \beta_{\mathrm{Age35-54}} \cdot \mathrm{Age}_{35-54}$ $+ \beta_{\text{Female}} \cdot \text{Female} + \beta_{\text{Diverse}} \cdot \text{Diverse}$ $+ \beta_{\text{HigherDegree}} \cdot \text{HigherDegree} + \beta_{\text{HighestDegree}} \cdot \text{HighestDegree}$ $+ \beta_{\text{HomeOwner}} \cdot \text{HomeOwner}$ + $\beta_{\text{IncomeBelow3,000}}$ · IncomeBelow3,000 + $\beta_{\text{NoIncomeGroup}}$ · NoIncomeGroup $+ \beta_{\text{TariffCompareNotAtAll}} \cdot \text{TariffCompareNotAtAll}$ $+ \beta_{\text{TechModeratelyInterested}} \cdot \text{TechModeratelyInterested} + \beta_{\text{TechVeryInterested}} \cdot \text{TechVeryInterested}$ $+ \beta_{ClimateModeratelyConcerned} \cdot ClimateModeratelyConcerned + \beta_{ClimateVeryConcerned} \cdot ClimateVeryConcerned$ $+ \beta_{\text{CarEmotional}} \cdot \text{CarEmotional} + \beta_{\text{CarNeutral}} \cdot \text{CarNeutral} + \beta_{\text{CarFunctional}} \cdot \text{CarFunctional} + \beta_{\text{CarRejecting}} \cdot \text{CarRejecting}$ $+ \beta_{\text{InfrequentDrivers}} \cdot \text{InfrequentDrivers} + \beta_{\text{DoNotDrive}} \cdot \text{DoNotDrive}$ $+ \beta_{\text{ShortDistances}} \cdot \text{ShortDistances} + \beta_{\text{LongDistances}} \cdot \text{LongDistances}$ $+ \beta_{\text{EVOwner}} \cdot \text{EVOwner} + \beta_{\text{EVNotInterested}} \cdot \text{EVNotInterested}$ $+ \beta_{\text{FamiliarWithV2G}} \cdot \text{FamiliarWithV2G} + \beta_{\text{VeryFamiliarWithV2G}} \cdot \text{VeryFamiliarWithV2G}$ $+ \beta_{\text{BatteryDegradation}} \cdot \text{BatteryDegradation} + \beta_{\text{Complexity}} \cdot \text{Complexity}$ $+ \beta_{\text{Duty}} \cdot \text{Duty} + \beta_{\text{RangeAnxiety}} \cdot \text{RangeAnxiety}$ $+ \beta_{\text{NudgeEnv}} \cdot \text{NudgeEnv}$ $+ \beta_{\text{SkimmedAttributes}} \cdot \text{SkimmedAttributes}$

 $+\beta_{\text{IncomeVarRem5ct/kWh}} \cdot \text{VarRem}_{5ct/kWh} \cdot \text{IncomeBelow3,000}$

 $+\beta_{\text{IncomeFixRem}} \cdot \text{FixRem} \cdot \text{IncomeBelow3,000}$

(1)

4. Results

This section presents the DCE's findings on the effects of selected contract attributes on the willingness to participate in V2G technology. First, a descriptive analysis is provided, followed by an assessment of the influence of contract attributes and nudging on participation probabilities using the MNL and MXL models. Finally, the willingness-to-accept (WTA) for various attributes is calculated to explore their economic implications.

4.1. Descriptive statistics

A total of 229 individuals participated in the survey. The dataset was cleaned following established procedures from Conrad et al. [49] and Leiner [50], excluding participants who completed the survey in less than 150 seconds. Furthermore, respondents who consistently selected "No participation in V2G" across all choice sets were excluded, as their responses offered no insights into the influence of contract attributes or nudging effects. After this filtering, 204 valid responses remained, comprising 100 participants who completed the ECON survey and 104 who completed the ENV survey.

Table 3 summarizes the socio-demographic characteristics of the participants that completed the ECON and ENV surveys, respectively. The final column includes average demographic data for Germany (2021–2022) to offer a benchmark for evaluating sample representativeness. Given the limited sample size for certain queries, some variables were grouped into broader categories to facilitate the analysis. The ungrouped socio-demographic statistics are detailed in Table D.1 in Appendix D.

Table 3: Socio-demographic characteristics of the sample and comparison with the German population

Category	Specification	ECON % (N=100)	ENV % (N=104)	Germany %
Age	18 to 34 years	41.00	45.19	24.72 [51]
	35 to 54 years	30.00	30.77	21.49 [51]
	55 years or older	29.00	24.04	53.79 [<mark>51</mark>]
Gender	Female	38.00	34.62	48.65 [51]
	Male	59.00	62.50	51.35 [<mark>51</mark>]
	Diverse	2.00	2.88	-
	No response	1.00	-	-
Level of education	School degree	5.00	4.81	18.3 [52]
	Higher degree ¹	80.00	81.73	70.2 [52]
	Highest degree ²	13.00	11.54	11.1 [52]
	No response	2.00	1.92	-
Housing situation	Rent	51.00	47.12	58.2 [53]
	Ownership	47.00	50.96	41.8 [53]
	No response	2.00	1.92	-
Income	Below €3,000 net	51.00	50.96	70.02 [54]
	€3,000 net or above	42.00	37.50	15.37 [54]
	No response	7.00	11.54	0.21^3 [54]

¹ Includes bachelor's and master's degrees and vocational training.

 2 Includes doctoral degrees.

³ Individuals without any salary were not considered.

Overall, socio-demographic distributions between the ECON and ENV groups are similar, with differences typically below 4%-points. However, certain deviations from national averages are notable. The sample is younger than the general population, which likely correlates with tech-savyness. Male participants were overrepresented compared to the general population, a trend often observed for surveys related to energy and EVs [55] [56]. While participants with higher degrees align closely with national averages, those with only school-level qualifications are underrepresented. The average proportion of renters in Germany is substantially higher. Related to this, the share of participants reporting monthly net incomes exceeding €3,000 is markedly above the national average. This is not surprising, as other studies report that current EV drivers have relatively high incomes [56]. While not fully representative of the general population, we argue

that the sample offers valuable insights into a key target group for V2G contracts: individuals with driver's licenses and interest in V2G technologies.

Figure 1 illustrates participant responses to additional questions regarding EV ownership, car relationship, EV and technical interests, familiarity with V2G, concerns about V2G and climate change, car usage, and habits to compare electricity tariffs. Detailed numbers are presented in Table D.2 in Appendix D.



Figure 1: Participant responses to additional questions

Note: The reference levels are marked in bold letters.

Notably, the groups responding to the ECON and ENV surveys are similar across most characteristics, indicating a successfully randomized distribution. Nevertheless, participants in the ENV survey are relatively more familiar with V2G, have a higher technical interest, and more frequently check their electricity tariffs. These factors should be kept in mind when evaluating the effect of nudging. For the other factors, most participants stated having a functional relationship with their car and being interested in EVs, while only relatively few own one. They are most often very or moderately concerned about climate change. V2G concerns are relatively rare among participants, with battery degradation being the most common. Participants typically travel short distances and are evenly distributed between frequent and infrequent drivers.

4.2. Determinants of the willingness to participate in V2G systems

This section examines the influence of contract attributes, participant characteristics, and environmental nudging on willingness to participate in V2G systems. The results of the MNL and MXL models are summarized in Table 4. It includes the estimated coefficients of the utility function (denoted by β), robust standard errors (denoted by SE), and robust significant levels (denoted by Sig.) for all contract attributes and those participant characteristics that yielded significance levels below 0.05. Sensitivity analyses were conducted to assess the robustness of the results across various model configurations for both MNL and MXL models. The analyses included varying the order and combinations of attributes, testing different functional forms (e.g., linear, quadratic), using different starting values for the MXL model, exploring various distributional assumptions, and incorporating interaction variables. These analyses confirmed the consistency of model outcomes, demonstrating the models' stability across different specifications and robustness to assumptions changes. Additional results are included in Appendix F.

			MNL Model			MXL Model			
Category	Reference	Variable	β	\mathbf{SE}	Sig.	β	\mathbf{SE}	Sig.	
Fixed remuneration	-	Linear 0-50 cents/day	0.017	0.002	***	0.033	0.003	***	
Variable remuneration	15 cents/kWh	10 cents/kWh	-0.078	0.089		-0.159	0.156		
		5 cents/kWh	-0.765	0.084	***	-0.780	0.202	***	
Contract flexibility	Daily	Monthly	-0.374	0.050	***	-0.962	0.156	***	
Minimum plug-in duration	5 hours	10 hours	-0.212	0.058	***	-0.125	0.143	*	
		15 hours	-0.889	0.063	***	-2.001	0.274	***	
Minimum battery level	70%	50%	-0.232	0.060	***	0.048	0.159		
		35%	-0.204	0.063	**	-0.586	0.174	***	
Income	€3,000 or above	Below €3,000	0.519	0.191	**	0.528	0.166	**	
		No income group	-0.435	0.205	*	-0.376	0.214		
Comparing electricity tariffs	Compare	Not compare	0.390	0.132	**	0.425	0.143	**	
Car relationship	No	Neutral	0.777	0.216	***	0.944	0.244	***	
		Rejecting	1.232	0.413	**	1.856	0.382	***	
EV ownership	Interested	Owner	-0.700	0.154	***	-0.782	0.172	***	
		Not interested in EV	-1.535	0.221	***	-1.724	0.248	***	
V2G familiarity	Not familiar	Very familiar	0.470	0.203	*	0.473	0.211	*	
V2G concerns	No	Battery degradation	-0.491	0.197	*	-0.426	0.224	*	
		Complexity	-0.690	0.273	*	-1.096	0.346	*	
		Range anxiety	-0.685	0.285	*	-0.853	0.328	**	
Environmental nudging ¹	No	Yes	0.473	0.145	**	0.465	0.161	**	
Interaction effects									
Income $< €3,000$		5 cents/kWh	0.374	0.116	**				
Income $< C3,000$		Fixed remuneration	0.005	0.002	*				
Model fit			Value			Value			
\mathbb{R}^2			0.256						
AIC			5092			5009			
BIC			5369			5305			

Table 4: Results of the estimation

Significance levels: *** p<0.001, ** p<0.01, * p<0.05.

 1 For calculating this coefficient, all the participants who reported not watching the video or reading the text, or spent less than 30 seconds on the corresponding page, were excluded.

Remuneration structure: Our analysis of the novel combination of fixed and variable remuneration reveals that both components significantly influence the willingness to participate. As expected, increasing fixed remuneration positively influences the willingness to participate, as reflected in both the MNL and MXL models. Similarly, decreasing variable remuneration negatively impacts the likelihood of V2G participation in both models. However, the estimated effect of reducing the variable remuneration from 15 to 10 cents per kWh is non-significant and disproportionately smaller than that of reducing the variable remuneration to 5 cents per kWh. This pattern suggests diminishing marginal utility, where increases in remuneration become less effective beyond a certain threshold. A range between 5 and 10 cents per kWh appears critical to making participation attractive. While the MNL model estimates similar effect sizes for changing fixed and variable remuneration from the best to the worst tested attribute level ($50 * 0.017 = 0.85 \simeq 0.765$), the MXL model attributes more weight to changing the fixed remuneration (50 * 0.033 = 1.65 > 0.780). Nevertheless, both models emphasize the importance of both fixed and variable incentives in encouraging V2G engagement.

Contract flexibility: Reducing contract flexibility from daily to monthly adjustability negatively significantly decreases willingness to participate in both models. This confirms our hypothesis on participants' aversion to long-term commitments. While the MNL model suggests that the effect size of changing the contract flexibility from daily to monthly is somewhat smaller than that of the maximum considered change in fixed and variable remuneration, the MXL model ranks the effect size of the considered change in contract flexibility between those of fixed and variable remuneration. While this finding is consistent with previous studies finding preferences for shorter contract durations (e.g., one month contract duration is preferred over 48 months), it reveals the additional preference for greater flexibility even within shorter contract durations. Furthermore, one hypothesis was that contract flexibility might also increase the willingness for longer connection times or lower battery levels, as these constraints could then be adjusted on a short-term basis. However, no interaction effects were observed.

Minimum plug-in duration: As expected, longer minimum plug-in durations significantly reduce the willingness to participate in V2G programs. The results for a 10-hour duration are consistent across models, with relatively small coefficients of -0.212 (MNL) and -0.125 (MXL). These values are modest compared to those associated with other contract attributes, indicating a less pronounced influence on participants' willingness to participate. In contrast, the 15-hour duration exhibits notably larger negative coefficients, especially in the MXL model. This difference likely reflects the MXL model's capacity to account for random taste variation, capturing participants' stronger aversion to longer plug-in durations. The non-linear relationship suggests a steep increase in negative preferences between 10 and 15 hours, pointing to a potential threshold many participants find difficult to accept.

Minimum battery level: Reducing the minimum battery level yields mixed results. On the one hand, the MNL model suggests that reducing the minimum battery level from 70% to 50% or 35% has significant similarly large negative effects on V2G participation. On the other hand, the MXL model shows no significant coefficient for the 50% level, highlighting considerable heterogeneity in participant preferences. While some participants may strongly dislike a 50% minimum, others may find it acceptable. The negative effect of the 35% level is more pronounced in the MXL model, suggesting that individuals with strong preferences impose a substantial penalty on lower battery levels—a nuance that the MNL model smooths out under its homogeneity assumption. Still, the coefficients of minimum battery levels remain relatively small compared to those of other contract attributes.

Electricity tariffs: Participants who do not actively compare electricity tariffs demonstrate a significantly higher likelihood of participating in V2G programs in both models. This result may appear counterintuitive, as one could expect participants who make an effort to compare electricity tariffs to save money on electricity bills would also be more inclined to make an effort to participate in V2G to benefit financially. Further research may be needed to resolve this apparent contradiction.

Car relationship: A neutral relationship with cars is associated with a significantly higher likelihood of participation in both models. Surprisingly, a rejecting relationship with cars also positively influences participation. This could stem from environmental concerns, where the benefits of V2G make car usage more acceptable, or from infrequent car usage, where V2G provides an opportunity to utilize parked vehicles.

EV ownership: EV ownership significantly reduces the likelihood of participation in V2G programs compared to individuals who do not own an EV but express a general interest in them. This result holds across both models. We can think of two potential reasons for this result. First, EV owners may be reluctant to replace their current charging technology, which likely does not support V2G. Second, EV owners may weigh more on V2G concerns when deciding on V2G participation. Those uninterested in EVs are also less likely to engage in V2G, likely because they perceive a lack of relevance in the absence of EV ownership.

V2G familiarity: High familiarity with V2G significantly increases participants' willingness to engage, as shown in both models. This finding aligns with research by Kester et al. [57], which suggests that knowledge about V2G systems enhances their acceptance.

V2G concerns: Concerns about battery degradation, process complexity, and range anxiety each negatively and significantly affect willingness to participate. Addressing these concerns through targeted information and marketing strategies could play a critical role in increasing acceptance and participation.

Nudging effect: Environmental nudging demonstrates a significant and positive effect on the willingness to participate in both models. Participants exposed to the ENV survey, which emphasized environmental benefits in the introductory video and through green highlighting, are more inclined to engage in V2G than those exposed to the ECON survey. The size of the nudging effect is within the range of previously discussed contract attributes, which underlines its relevance. Some participants reported not watching their assigned video or spent insufficient time on the introduction page. Still, we cannot distinguish the effect of the video from that of the green highlighting due to the small sample size. For our main results, participants who reported not watching their assigned video or spent insufficient time on the introductory page were excluded, but a sensitivity run without this exclusion yielded similar results. Notably, this nudging effect is significant after controlling for participant characteristics like V2G familiarity and technical interest. Hence, we eliminate the risk that our finding on nudging is biased by these characteristics not being perfectly balanced between the ENV and ECON respondents. Furthermore, interaction terms between these characteristics were non-significant, suggesting that the effect of nudging does not depend on prior traits.

Interaction effects: The interaction effect between a lower income and a variable remuneration of 5 cents per kWh is significant and positive. This indicates that lower-income participants are more sensitive to this remuneration scheme than higher-income participants, with a stronger preference for even small compensation amounts. In contrast, the difference in participation between 10 and 15 cents per kWh for lower-income participants is non-significant, suggesting that once a higher remuneration is offered, lower-income individuals may become more indifferent to further increases in variable remuneration.

Similarly, the interaction effect between lower income and fixed remuneration is also significant and positive, indicating a higher value placed on fixed payments by lower-income participants. This may be because fixed remuneration provides certainty, unlike variable payments that depend on usage. This assumption is supported by Holt and Laury [58], who found that income is negatively correlated with risk aversion, meaning lower-income individuals may be more risk-averse and place less importance on uncertain remuneration per kWh. Furthermore, for higher-income groups, the value of earnings from V2G participation is relatively small compared to their total income, potentially leading them to pay less attention to them.

Non-significant parameters: We do not find significant effects of age, gender, education, concerns about climate change, a functional or emotional relationship with cars, and driving behavior on the willingness to participate in V2G. Additionally, the sample sizes for participants expressing concerns about digitization, data sharing, financial benefit doubts, or external interference were too small for statistical evaluation.

4.3. Willingness to accept changes in contract attributes

This section translates the previous estimates into the WTA for a deterioration in the different V2G contract attributes. Given the previously found significant interaction terms between income and remuneration, we calculate distinct WTAs for participants earning below and above &3,000 per month net. Participants who did not disclose income information were excluded due to the small sample size of this subgroup.

To this end, the parameters were re-estimated for the different income groups using the MNL model exclusively. The decision to omit the MXL model is justified by the similarity in parameter estimates between the MNL and MXL models, suggesting that the MXL model's additional complexity is unnecessary for the WTA analysis. Furthermore, we excluded the 15-hour minimum plug-in duration from the WTA analysis due to the previously found strong negative effect suggesting the impracticality of such long plug-in durations. The results are summarized in Appendix G.

The WTA for each significant contract attribute was calculated as the negative ratio of the attribute's coefficient to the fixed remuneration coefficient, representing the marginal rate of substitution:

$$WTA = -\frac{\beta_a}{\beta_{FixRem}},\tag{2}$$

where β_a denotes the coefficient of the attribute of interest and β_{FixRem} represents the coefficient of fixed remuneration. This quotient indicates the additional income required to compensate for the negative effects of an attribute's deterioration.

The results are summarized in Table 5 and reveal that participants in the lower-income group consistently exhibit lower WTA values than their higher-income counterparts. This suggests that lower-income individuals attribute greater importance to fixed income and are willing to accept reductions in variable remuneration and comfort for comparatively smaller increments in fixed remuneration.

		$\mathbf{Income} < \textbf{€3,000}$	$\textbf{Income} \geq \textbf{€3,000}$
Attribute	Attribute change	Cents/day WTA	Cents/day WTA
Variable remuneration	$15~{\rm cents/kWh}$ to $5~{\rm cents/kWh}$	17.26	51.08
Contract flexibility	Daily to monthly	15.54	26.95
Minimum plug-in duration	5 hours to 10 hours	8.74	/
Minimum battery level	70% to $35%$	/	22.47
Minimum battery level	70% to $50%$	9.45	17.31

Table 5: Willingness to accept relative to fixed remuneration

Variable remuneration: Reducing variable remuneration from 15 to 5 cents per kWh would require an increase in fixed remuneration of 17.26 cents per day for the lower-income group and 51.08 cents per day for the higher-income group to accept the change. This demonstrates that the higher-income group is significantly more sensitive to changes in variable remuneration. This finding aligns with previous research suggesting that income negatively correlates with risk aversion [58, e.g.]. In other words, the lower the income, the higher the risk aversion and the higher the value from additional income. In the context of this experiment, this means that individuals pay a higher importance to the fixed remuneration per day.

Contract flexibility: The lower-income group places a value of 15.54 cents per day on transitioning from daily to monthly contracts, whereas the higher-income group assigns a value of 26.95 cents per day to the same change. This results indicates that both groups value contract flexibility. Furthermore, it questions the future relevance of monthly fixed contracts, as the benefits of longer fixed terms may not sufficiently offset the required increase in financial incentives.

Minimum plug-in duration: The lower-income group requires an additional 8.73 cents per day to accept an increase in the plug-in duration from 5 to 10 hours. This change has no significant impact on the higherincome group. Conversely, for the higher-income group, the lack of significant effect may stem from the fact that people with a higher income use EVs more often as a second vehicle [59]. Thus, the minimal plug-in duration might be less important for them.

Minimum battery level: For the higher-income group, reducing the minimum battery level from 70% to 35% is valued at 22.47 cents per day, while a reduction from 70% to 50% is worth 17.31 cents per day. In contrast, the same reduction from 70% to 50% is only valued at 9.45 cents per day by the lower-income group. The reduction from 70% to 35% is only significant in the high-income group. This might be due to the fact that people with a higher income tend to drive more kilometers on average [60], hence this group exhibits greater sensitivity to reductions in the minimum battery level.

The WTA analysis reveals distinct differences in how income groups value the selected contract attributes. Participants with a net income below \pounds 3,000 per month prioritize fixed remuneration and can be more easily incentivized to alter their behavior through fixed payments. Conversely, participants earning \pounds 3,000 or more per month place greater importance on variable remuneration, contract flexibility, and minimum battery level. These findings underscore the existence of two distinct target groups, which can be addressed through tailored strategies to enhance grid utility effectively.

5. Discussion

This section contextualizes the findings of our study within the existing literature and discusses limitations. Four main distinctions are identified: we examine the combination of fixed and variable remuneration, introduce and test the concept of contract flexibility, update previous findings on further contract attributes like minimum battery levels, and investigate the effect of environmental nudging. Not least, we differentiate our findings by income groups.

While previous studies only considered fixed remuneration (see Appendix H), we find that both fixed and variable remuneration significantly affect the willingness to participate. The two remuneration components yield the highest WTA among our considered contract attributes and levels across income groups. However,

our results suggest that lower-income participants prefer fixed remuneration, and those with higher incomes favor variable remuneration.⁴

Without previous research on sub-monthly contract flexibility, we can use earlier studies that investigated contract durations between six and 48 months for comparison (see Appendix H). Our results suggest that increasing contract flexibility from monthly to daily significantly increases the willingness to participate in V2G systems. Contract flexibility even yields the third-highest WTA among our considered contract attributes—less than the remuneration components but more than the minimum plug-in duration and battery level. This finding differs from Kubli et al. [26], who find that contract duration is less important than minimum battery levels. The higher importance of contract flexibility might be explained by a decreasing marginal effect of commitment duration, be it short-term contract flexibility or more long-term contract duration. Supporting this notion, Huang et al. [27] show minimal differentiation in participant preferences between 6- and 24-month contracts.

Furthermore, we update previous findings on other contract attributes, namely minimum plug-in duration and battery level, and differentiate these findings by income. On the one hand, previous studies suggest greater importance of minimum plug-in durations and minimum battery levels than our results (see Appendix H). These differences could be attributed to technological progress and model mis-specifications. Technological progress has enabled rapidly increasing EV battery capacities over the last few years, which could explain the reducing relevance of minimum battery levels, particularly when compared to older studies [24, 25]. Moreover, the choice of minimum battery levels seems crucial, as the levels considered in some studies would not be sufficient for emergencies or could damage the battery, thereby fostering a strong negative effect on the entire attribute [25, 27]. Finally, the same studies did not dummy-code the plug-in durations [25, 27]. This can potentially distort the results by failing to capture preference thresholds, like the one we identified between 10 and 15 hours minimum plug-in duration.

The significant impact of environmental nudging, which operates independently of the contract attributes, represents a novel finding in the field of V2G. Still, our result aligns well with previous studies finding positive effects for other types of pro-environmental behavior. For instance, Wee et al. [29] analyzed 37 articles and found that nudging was effective in 64.86% of cases. Our research supports the hypothesis that environmental nudging is also effective for V2G and quantifies the effect relative to other contract attributes.

 $^{^{4}}$ Note that Huang et al. [27] also discuss a "variable" remuneration component, but this rewards participants for additional plug-in time as opposed to the V2G activation in our case.

When interpreting the results of our study, two main limitations should be kept in mind. First, the reliance on stated preference data introduces inherent limitations. Participants are not exposed to the real-world consequences of their choices, which may lead to underestimating the importance of certain attributes or overstating hypothetical preferences. Nevertheless, in the context of V2G, stated preferences offer a valuable preview of consumer preferences before the new technology is deployed at scale. Second, while the sample size of 204 participants is sufficiently large to derive statistically significant results, generalizability is naturally limited. From the demographic characteristics of our sample, we can conclude that our results are not representative of the German population. However, through dissemination channels like EV internet forums, we may have reached survey participants who are also more likely to use V2G systems, which could also increase the relevance of our results.

6. Conclusion

In the context of transport electrification, V2G technology can potentially support balancing the electricity systems. However, the successful implementation of V2G depends significantly on understanding the willingness of EV owners to participate and designing effective contracts to facilitate engagement. This study contributes to this understanding by investigating the willingness of current and potential future German EV drivers to participate in V2G through a DCE.

We identify the effects of specific contract attributes, participant characteristics, and environmental nudging on the willingness of EV owners to participate in V2G. Besides fixed remuneration, we find that the newly investigated contract attributes of variable remuneration and contract flexibility positively and significantly influence participation. Conversely, attributes such as a lower guaranteed battery level and high minimal plug-in duration negatively impacted willingness to participate, especially when thresholds were exceeded. In addition to contract attributes, our study highlights the great potential of environmental nudging in increasing participation rates. Interestingly, we find that those participants who currently own an EV are less likely to participate in V2G programs than individuals who report being interested in owning an EV in the future.

We also identified income-specific preferences. Participants with a net income below C3,000 per month prioritized fixed remuneration, the minimum plug-in duration significantly influenced participation, but the minimum battery level was insignificant. In contrast, participants with a net income of C3,000 or more placed greater importance on variable remuneration and considered minimum battery level more critical than plug-in duration.

Several practical recommendations can be drawn from our results. For businesses, our differentiated findings on V2G preferences underscore the value of customizing contract designs and marketing efforts to address the unique needs of various user segments, particularly those with different incomes. Furthermore, companies may consider a hybrid remuneration structure to address diverse preferences, offer flexible contract durations to accommodate consumer preferences for spontaneity, and leverage our positive results on nudging by emphasizing environmental benefits in their marketing efforts. To address challenges in meeting plug-in duration requirements, businesses may explore solutions that do not necessitate prolonged home parking. Additionally, setting default minimum battery levels at 35% may prevent participants indifferent to battery levels from opting for higher thresholds, thus optimizing the grid's usable capacity.

Policymakers may also play a role in improving V2G familiarity and raising awareness of V2G's environmental benefits. Information campaigns and public education could emphasize how V2G supports grid stability, integrates RES, and reduces environmental impact. Policymakers may also support research and pilot projects to collect further data on user preferences, refine contract designs, and eventually promote broader V2G participation.

Further research could build on and expand our research in several relevant areas. First, to improve the design efficiency of DCEs, future studies could incorporate the coefficients identified in this research as prior estimates. Adjustments, such as reducing the number of attributes, alternatives, or choice sets, could also enhance efficiency. For more granular insights, attribute levels could be tested with finer increments, and differences between weekday and weekend behaviors could be explored. Second, as pilot projects expand, future studies could employ revealed preference methods to validate and refine the findings from studies based on stated preferences like ours. Third, future research may conduct further segmented analyses, which could enable V2G offerings that are more precisely tailored to specific user groups. In particular, our findings encourage a more detailed analysis of the potential impact of variable remuneration adjustments on higher-income groups, as these changes could be highly influential.

Acknowledgements

The authors would like to thank all survey participants. Furthermore, we thank Marc Oliver Bettzüge, Paul Fabianek, Stefan Poier, and participants of the joint ewi-FCN Research Seminar in Energy Economics of the University of Cologne and RWTH Aachen for thoughtful and constructive comments and discussions on this work.

Funding: This work was supported by the Ministry of Economic Affairs, Industry, Climate Protection and Energy of the State of North Rhine-Westphalia (MWIKE) within the project 'VISE - Smart: Data Generating added value through energy data - trends and transformation processes' (grant number EFO 0151D).

References

- [1] International Energy Agency (IEA), Global EV Data Explorer, Retrieved November 19, 2024, from https://www.iea.org/data-and-statistics/data-tools/global-ev-data-explorer (2024).
- [2] C. Liu, K. T. Chau, D. Wu, S. Gao, Opportunities and challenges of vehicle-to-home, vehicle-tovehicle, and vehicle-to-grid technologies, Proceedings of the IEEE 101 (11) (2013) 2409-2427. doi: 10.1109/JPROC.2013.2271951.
- [3] W. Kempton, S. E. Letendre, Electric vehicles as a new power source for electric utilities, Transportation Research Part D: Transport and Environment 2 (3) (1997) 157–175. doi:10.1016/S1361-9209(97)00001-1.
- [4] C. Nobis, T. Kuhnimhof, Mobilität in deutschland mid ergebnisbericht, Retrieved January 12, 2024, from https://elib.dlr.de/125879/1/MiD2017_Ergebnisbericht.pdf (2018).
- [5] J. Figgener, C. Hecht, D. Haberschusz, J. Bors, K. G. Spreuer, K.-P. Kairies, P. Stenzel, D. U. Sauer, The development of battery storage systems in germany: A market review (status 2023) (2022). doi: 10.48550/arXiv.2203.06762.
- [6] C. G. Hoehne, M. V. Chester, Optimizing plug-in electric vehicle and vehicle-to-grid charge scheduling to minimize carbon emissions, Energy 115 (2016) 646–657. doi:10.1016/j.energy.2016.09.057.
- [7] J. Dixon, W. Bukhsh, K. Bell, C. Brand, Vehicle to grid: driver plug-in patterns, their impact on the cost and carbon of charging, and implications for system flexibility, eTransportation 13 (2022) 100180. doi:10.1016/j.etran.2022.100180.
- [8] M. A. Hannan, M. S. Mollik, A. Q. Al-Shetwi, S. A. Rahman, M. Mansor, R. A. Begum, K. M. Muttaqi, Z. Y. Dong, Vehicle to grid connected technologies and charging strategies: Operation, control, issues and recommendations, Journal of Cleaner Production 339 (2022) 130587. doi: 10.1016/j.jclepro.2022.130587.
- [9] Renault, Renault 5 e-tech electric: das neue gesicht der elektrischen revolution, Retrieved May 7, 2024, from https://presse.renault.de/renault-5-e-tech-electric-das-neue-gesicht-der-elektrischen-revolution/ (2024).
- [10] B. Tiemann, Bidirektionales laden, Retrieved November 26, 2023, from https://www.beckertiemann.de/bidirektionales-laden/ (2023).
- [11] Honda, Vehicle-to-grid-technologie: Honda schließt machbarkeitsprojekt erfolgreich ab, Retrieved November 26, 2023, from https://hondanews.eu/ch/de/corporate/media/pressreleases/430111/ vehicle-to-grid-technologie-honda-schliesst-machbarkeitsprojekt-erfolgreich-ab (2023).
- [12] M.-B. Mobility, Charging solutions, Retrieved November 26, 2023, from https://www.mercedes-benzmobility.com/de/was-wir-tun/charging/ (2023).
- [13] European Union Agency for the Cooperation of Energy Regulators (ACER) and Council of European Energy Regulators (CEER), Energy retail - Active consumer participation is key to driving the energy transition: how can it happen? 2024 Market Monitoring Report, https://www.acer.europa.eu/sites/ default/files/documents/Publications/ACER-CEER_2024_MMR_Retail.pdf (2024).
- T. Signer, N. Baumgartner, M. Ruppert, T. Sandmeier, W. Fichtner, Modeling v2g spot market trading: The impact of charging tariffs on economic viability, Energy Policy 189 (2024) 114109. doi:https: //doi.org/10.1016/j.enpol.2024.114109.

URL https://www.sciencedirect.com/science/article/pii/S0301421524001290

- [15] C. Hecht, J. Figgener, D. U. Sauer, Vehicle-to-grid market readiness in europe with a special focus on germany, Vehicles 5 (4) (2023) 1452–1466. doi:10.3390/vehicles5040079.
- [16] B. K. Sovacool, L. Noel, J. Axsen, W. Kempton, The neglected social dimensions to a vehicle-to-grid (v2g) transition: a critical and systematic review, Environmental Research Letters 13 (1) (2018) 013001. doi:10.1088/1748-9326/aa9c6d.
- [17] H. K. Park Lee, A socio-technical exploration of the car as power plant, Ph.D. thesis, Delft University of Technology (2019). doi:10.4233/uuid:bcf14d54-74cc-4fe8-9285-466cee3936ab.
- [18] Everoze EVConsult, V2g global roadtrip: around the world in 50 projects: lessons learned from fifty international vehicle-to-grid projects, Retrieved February 5, 2024, from https://everoze.com/app/ uploads/2019/02/UKPN001-S-01-J-V2G-global-review.pdf (2018).

- [19] Y. Gao, Y. Chen, C.-Y. Wang, K. J. Ray Liu, A Contract-Based Approach for Ancillary Services in V2G Networks: Optimality and Learning, IEEE, 2013.
- [20] A. G. Jember, W. Xu, C. Pan, X. Zhao, X.-C. Ren, Game and contract theory-based energy transaction management for internet of electric vehicle, IEEE Access 8 (2020) 203478–203487. doi:10.1109/ ACCESS.2020.3036415.
- [21] K. Zhang, Y. Mao, S. Leng, M. Zeng, L. Xu, L. Jiang, A. Vinel, Optimal energy exchange schemes in smart grid networks: A contract theoretic approach, in: 2016 IEEE/CIC International Conference on Communications in China (ICCC), IEEE, 2016, pp. 1–6. doi:10.1109/ICCChina.2016.7636834.
- [22] C. Guille, G. Gross, A conceptual framework for the vehicle-to-grid (v2g) implementation, Energy Policy 37 (11) (2009) 4379–4390. doi:10.1016/j.enpol.2009.05.053.
- [23] Z. Jiao, Y. Yin, L. Ran, Z. Gao, Integrating vehicle-to-grid contract design with power dispatching optimisation: managerial insights, and carbon footprints mitigation, International Journal of Production Research 60 (17) (2022) 5354–5379. doi:10.1080/00207543.2021.1956694.
- [24] G. R. Parsons, M. K. Hidrue, W. Kempton, M. P. Gardner, Willingness to pay for vehicle-to-grid (v2g) electric vehicles and their contract terms, Energy Economics 42 (2014) 313–324. doi:10.1016/ j.eneco.2013.12.018.
- [25] J. Geske, D. Schumann, Willing to participate in vehicle-to-grid (v2g)? why not!, Energy Policy 120 (2018) 392-401. doi:10.1016/j.enpol.2018.05.004.
- [26] M. Kubli, M. Loock, R. Wüstenhagen, The flexible prosumer: Measuring the willingness to co-create distributed flexibility, Energy Policy 114 (2018) 540–548. doi:10.1016/j.enpol.2017.12.044.
- [27] B. Huang, A. G. Meijssen, J. A. Annema, Z. Lukszo, Are electric vehicle drivers willing to participate in vehicle-to-grid contracts? a context-dependent stated choice experiment, Energy Policy 156 (2021) 112410. doi:10.1016/j.enpol.2021.112410.
- [28] A. Palm, Early adopters and their motives: Differences between earlier and later adopters of residential solar photovoltaics, Renewable and Sustainable Energy Reviews 133 (2020) 110142. doi:10.1016/ j.rser.2020.110142.
- [29] S.-C. Wee, W.-W. Choong, S.-T. Low, Can "nudging" play a role to promote pro-environmental behaviour?, Environmental Challenges 5 (2021) 100364. doi:10.1016/j.envc.2021.100364.
- [30] S. Saxena, H. E. Z. Farag, L. St. Hilaire, A. Brookson, A techno-social approach to unlocking vehicle to everything (v2x) integration: A real-world demonstration, IEEE Access 11 (2023) 17085–17095. doi:10.1109/ACCESS.2023.3244562.
- [31] P. P. Malya, L. Fiorini, M. Rouhani, M. Aiello, Electric vehicles as distribution grid batteries: a reality check, Energy Informatics 4 (S2) (2021). doi:10.1186/s42162-021-00159-3.
- [32] H. Moon, S. Y. Park, C. Jeong, J. Lee, Forecasting electricity demand of electric vehicles by analyzing consumers' charging patterns, Transportation Research Part D: Transport and Environment 62 (2018) 64–79. doi:10.1016/j.trd.2018.02.009.
- [33] ADAC, Weite wege zur notdienst-apotheke, Retrieved January 12, 2024, from https://www.adac.de/ gesundheit/gesund-unterwegs/vorsorge/apotheken-notdienst/ (2021).
- [34] D. Yang, H. Liu, M. Li, H. Xu, Data-driven analysis of battery electric vehicle energy consumption under real-world temperature conditions, Journal of Energy Storage 72 (2023) 108590. doi:10.1016/ j.est.2023.108590.
- [35] E. D. Kostopoulos, G. C. Spyropoulos, J. K. Kaldellis, Real-world study for the optimal charging of electric vehicles, Energy Reports 6 (2020) 418–426. doi:10.1016/j.egyr.2019.12.008.
- [36] J. J. Louviere, D. Street, L. Burgess, N. Wasi, T. Islam, A. A. Marley, Modeling the choices of individual decision-makers by combining efficient choice experiment designs with extra preference information, Journal of Choice Modelling 1 (1) (2008) 128–164. doi:10.1016/S1755-5345(13)70025-3.
- [37] ChoiceMetrics, Ngene 1.2 user manual & reference guide: The cutting edge in experimental design, Retrieved January 5, 2024, from https://choice-metrics.com/NgeneManual120.pdf (2018).
- [38] M. Bech, T. Kjaer, J. Lauridsen, Does the number of choice sets matter? results from a web survey applying a discrete choice experiment, Health economics 20 (3) (2011) 273–286. doi:10.1002/hec.1587.
- [39] D. McFadden, Conditional logit analysis of qualitative choice behavior, in: Frontiers in econometrics, 1974, pp. 105–142.

- [40] K. E. Train, Discrete Choice Methods with Simulation, Cambridge University Press, 2012. doi: 10.1017/CB09780511805271.
- [41] D. A. Hensher, W. H. Greene, The mixed logit model: The state of practice, Transportation 30 (2) (2003) 133-176. doi:10.1023/A:1022558715350.
- [42] J. J. Louviere, Choice experiments: an overview of concepts and issues, in: The Strengths and Weaknesses of Environmental Choice Modelling, Edward Elgar Publishing, 2001, pp. 13–36. doi: 10.4337/9781781956601.00010.
- [43] E. Lancsar, D. G. Fiebig, A. R. Hole, Discrete choice experiments: A guide to model specification, estimation and software, PharmacoEconomics 35 (7) (2017) 697–716. doi:10.1007/s40273-017-0506-4.
- [44] M. Bierlaire, A short introduction to pandasbiogeme, Retrieved March 13, 2024, from http:// infoscience.epfl.ch/record/283453 (2020).
- [45] M. Bierlaire, Mixture of logit models, Retrieved March 13, 2024, from https://biogeme.epfl.ch/ sphinx/auto_examples/swissmetro/plot_b06unif_mixture_integral.html#sphx-glr-autoexamples-swissmetro-plot-b06unif-mixture-integral-py (2023).
- [46] A. Alin, Multicollinearity, WIREs Computational Statistics 2 (3) (2010) 370–374. doi:10.1002/ wics.84.
- [47] K. P. Burnham, D. R. Anderson, Multimodel inference, Sociological Methods & Research 33 (2) (2004) 261–304. doi:10.1177/0049124104268644.
- [48] P. Mariel, D. Hoyos, J. Meyerhoff, M. Czajkowski, T. Dekker, K. Glenk, J. B. Jacobsen, U. Liebe, S. B. Olsen, J. Sagebiel, M. Thiene, Environmental Valuation with Discrete Choice Experiments, Springer International Publishing, Cham, 2021. doi:10.1007/978-3-030-62669-3.
- [49] F. G. Conrad, M. P. Couper, R. Tourangeau, C. Zhang, Reducing speeding in web surveys by providing immediate feedback, Survey research methods 11 (1) (2017) 45-61. doi:10.18148/srm/ 2017.v11i1.6304.
- [50] D. J. Leiner, Too fast, too straight, too weird: Non-reactive indicators for meaningless data in internet surveys: 229-248 pages / survey research methods, vol 13 no 3 (2019) (2019). doi:10.18148/srm/ 2019.v13i3.7403.
- [51] Statistisches Bundesamt, Bevölkerung, erwerbstätige, erwerbslose, erwerbspersonen, nichterwerbspersonen aus hauptwohnsitzhaush.: Deutschland, jahre, geschlecht, altersgruppen, Retrieved May 20, 2024, from https://www-genesis.destatis.de/genesis//online?operation=table&code=12211-0001&bypass=true&levelindex=1&levelid=1715763841489#abreadcrumb (2022).
- [52] Statistisches Bundesamt, Bildung: Auszug aus dem datenreport 2021, Retrieved May 20, 2024, from https://www.destatis.de/DE/Service/Statistik-Campus/Datenreport/Downloads/datenreport-2021-kap-3.pdf?__blob=publicationFile (2021).
- [53] Statistisches Bundesamt, Eigentumsquote: Wohnen, Retrieved May 20, 2024, from https://www.destatis.de/DE/Themen/Gesellschaft-Umwelt/Wohnen/Tabellen/tabelleeigentumsquote.html (2022).
- [54] Statistisches Bundesamt, Bevölkerung, erwerbstätige, erwerbslose, erwerbspersonen, nichterwerbspersonen aus hauptwohnsitzhaush.: Deutschland, jahre, geschlecht, größenkl. persönl. monatl. nettoeinkommen, Retrieved May 20, 2024, from https://www-genesis.destatis.de/ genesis//online?operation=table&code=12211-0003&bypass=true&levelindex=1&levelid= 1715763841489#abreadcrumb (2022).
- [55] IRENA, Renewable energy and jobs: Annual review 2019, Retrieved May 15, 2024, from https: //www.irena.org/-/media/Files/IRENA/Agency/Publication/2019/Jun/IRENA_RE_Jobs_2019report.pdf?rev=58ac56eaa71242309b0278055747df68 (2019).
- [56] D. M. N. John E. Anderson, Moritz Bergfeld, F. Steck, Real-world charging behavior and preferences of electric vehicles users in germany, International Journal of Sustainable Transportation 17 (9) (2023) 1032–1046. doi:10.1080/15568318.2022.2147041.
- [57] J. Kester, L. Noel, G. Zarazua de Rubens, B. K. Sovacool, Promoting vehicle to grid (v2g) in the nordic region: Expert advice on policy mechanisms for accelerated diffusion, Energy Policy 116 (2018) 422-432. doi:10.1016/j.enpol.2018.02.024.

[58] C. A. H. Holt, S. K. Laury, Risk aversion and incentive effects, The American Economic Review 92 (5) (2002) 1644–1655.

URL http://www.jstor.org/stable/3083270

- [59] S. Skippon, M. Garwood, Responses to battery electric vehicles: Uk consumer attitudes and attributions of symbolic meaning following direct experience to reduce psychological distance, Transportation Research Part D: Transport and Environment 16 (7) (2011) 525–531. doi:10.1016/j.trd.2011.05.005.
- [60] M. E. Kahn, Do richer people pollute more or less? new evidence on household level vehicle emissions kuznets curves, SSRN Electronic Journal (2004). doi:10.2139/ssrn.564627.

Abbreviations

Table 6: Table	of abbrevi	ations
Akaike information criterion	MNL	Multinomial Logit
Aternative-specific constant	MXL	Mixed Logic
Bayesian information criterion	RES	Renewable energy source
Discrete choice experiment	SE	Standard errors
Economical	V2G	Mobility cluster
Environmental	VIF	Variance inflation factor
Electric vehicle	WTA	Willingness to accept
	Table 6: Table Akaike information criterion Aternative-specific constant Bayesian information criterion Discrete choice experiment Economical Environmental Electric vehicle	Table 6: Table of abbreviAkaike information criterionMNLAternative-specific constantMXLBayesian information criterionRESDiscrete choice experimentSEEconomicalV2GEnvironmentalVIFElectric vehicleWTA

Table 6: Table of abbreviations

Appendices

A. Design of choice sets

In this section, table A.1 presents the design produced by Ngene. It contains 15 choice sets, as indicated by the row "choice set". The rows of alternative 1 describe the first contract offered, whereas the columns of alternative 2 describe the second contract offered. They are divided in the different contract attributes. The numbers from 0 to 2 indicate the corresponding attribute level as defined in table A.2.

Table A.1: Design of choice sets															
Choice Set	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
A1: Fixed remuneration	2	0	0	2	2	1	1	2	1	2	1	0	1	0	0
A1: Variable remuneration	1	1	1	0	0	0	1	2	2	2	0	2	2	0	1
A1: Contract flexibility	1	0	1	0	0	1	1	0	0	1	1	1	0	0	0
A1: Min. plug-in duration	2	0	1	2	1	1	0	0	0	0	2	2	1	2	1
A1: Min. battery level	0	1	2	0	2	1	0	1	2	2	2	0	1	1	0
A2: Fixed remuneration	0	1	1	1	0	0	2	0	2	1	2	1	0	2	2
A2: Variable remuneration	0	2	0	2	2	1	2	0	1	1	2	1	0	1	0
A2: Contract flexibility	0	1	0	1	0	1	0	1	1	0	0	0	1	0	1
A2: Min. plug-in duration	1	2	0	1	0	2	2	1	2	2	1	0	0	1	0
A2: Min. battery level	1	0	0	2	1	2	2	0	1	1	0	2	2	0	1

Table A.1: Design of choice sets

Note: The abbreviation A1 and A2 correspond to the different alternatives of the DCE.

 Table A.2: Attributes and Attribute Levels

Level	Attribute	Attribute Level
0		$0 \mathrm{~cents/day}$
1	Fixed remuneration	$25 \ {\rm cents/day}$
2		$50 {\rm cents/day}$
0		5 cents/kWh
1	Variable remuneration	$10 \ \mathrm{cents/kWh}$
2		$15 \ \mathrm{cents/kWh}$
0	Change of contract	Daily
1		Monthly
0		5 hours
1	Plug-in duration	10 hours
2		15 hours
0		35%
1	Minimal battery level	50%
2		70%

B. Example of the choice sets

Erklärung der Vertragsbestandteile

	1. Vertrag	2. Vertrag	Keine Teilnahme an V2G
1. Mindestanschlusszeit	15 Stunden	10 Stunden	
2. Minimales Batterielevel während V2G	35 %	50 %	
3. Änderung der Vertragskonditionen	Monatlich möglich	Täglich möglich	
4. Variable finanzielle Entlohnung	10 Cent/kWh	5 Cent/kWh	
5. Fixe finanzielle Entlohnung	50 Cent/Tag	0 Cent/Tag	

1. Welchen Vertrag präferieren Sie?

Wenn keiner der Verträge Sie überzeugt, wählen Sie bitte "Keine Teilnahme an V2G"

1. Vertrag

2. Vertrag

Keine Teilnahme an V2G

Figure B.1: The choice sets as presented in the reference survey (ECON)

Erklärung der Vertragsbestandteile									
		1. Vertrag 🖉	2. Vertrag 🖉	Keine Teilnahme an V2G					
	1. Mindestanschlusszeit	15 Stunden	10 Stunden						
	2. Minimales Batterielevel während V2G	35 %	50 %						
	3. Änderung der Vertragskonditionen	Monatlich möglich	Täglich möglich						
	4. Variable finanzielle Entlohnung	10 Cent/kWh	5 Cent/kWh						
	5. Fixe finanzielle Entlohnung	50 Cent/Tag	0 Cent/Tag						

1. Welchen Vertrag präferieren Sie?

Wenn keiner der Verträge Sie überzeugt, wählen Sie bitte "Keine Teilnahme an V2G"

1. Vertrag

2. Vertrag

Keine Teilnahme an V2G

Figure B.2: The choice sets with the green-colored V2G contract options (ENV)

C. Survey questions

This section contains a list of all survey question categories followed by a list of all survey questions and answer options.

Question Categories:

- Age
- Gender
- Country
- Housing situation
- Educational attainment
- Net income
- Car relation
- Electricity tariff control
- Perceived threat from climate change
- Interest in technical topics
- V2G knowledge
- Concerns regarding V2G (Battery degradation, Range, Commitment obligation, Complex process, Digitization of car usage, Data privacy)
- Explanation of contract components was read
- Explanation of V2G was watched
- Frequency of car usage
- Average kilometers on a working day
- Electric car ownership

Questions:

- 1. How old are you? (18 to 24 years old; 25 to 34 years old; 35 to 44 years old; 45 to 54 years old; 55 to 64 years old; 65 years old or older)
- 2. What is your gender? (Female; Male; Diverse)
- 3. In which country are you living right now? (Germany; Austria; Switzerland; Another country)
- 4. What is your current living situation? (Renting an apartment; Renting a house; Owner of an apartment; Owner of a house; Other)
- 5. What is your highest level of education? (Currently a pupil; Finished school without a degree; School degree; Completed vocational training; University degree; Doctoral degree; Other degree)

- 6. What is approximately your monthly net income? (Less than €500; €500 to less than €1,000; €1,000 to less than €3,000; €3,000 to less than €5,000; €5,000 and more)
- 7. How would you describe your relationship to cars? (Emotional; Neutral; Functional; Rejecting; Other)
- 8. Have you ever checked if you could get better terms by changing your electricity contract? (Yes, regularly; Yes, occasionally; No; I don't take care of the electricity bill)
- 9. I perceive climate change as a threat (5 options ranging from 'not at all' to 'very'; 'climate change does not exist')
- 10. I have an interest in technical subjects (5 options ranging from 'not at all' to 'very')
- I was familiar with the Vehicle to Grid concept before this survey (5 options ranging from 'not at all' to 'very')
- 12. Did you read the explanation about the contract components? (Yes, thoroughly; Yes, skimmed; No)
- 13. Did you watch the explanation about V2G in this experiment? (Video and/or text form) (Yes; Yes, but stopped/skimmed; No)
- 14. Do you have concerns about using V2G? (Yes; No)
 - (a) If yes: what are your main concerns?
 - i. The battery will be weakened by the additional charging cycles
 - ii. The range might not be sufficient
 - iii. An obligation to connect arises
 - iv. The process is complex
 - v. The use of the car is too digitalized
 - vi. Data, e.g., about charging behavior, is shared
 - vii. Other
 - (b) Else: Nothing
- 15. How often do you use the car on a weekly average? (Several times a day; Once a day; 3 to less than 7 times a week; 1 to less than 3 times a week; Not at all)
- 16. How many kilometers do you usually drive on an average weekday? (0 km; 1 km to less than 50 km; 50 km to less than 200 km; more than 200 km)
- 17. Do you own an electric car? (Yes (owned/leased); No, but I plan to buy one; No, I don't plan to buy one but do not exclude it for the future; No, and I do not want one; Other)

D. Socio-demographic and characteristic values

This section contains a detailed listing of the socio-demographic responses in absolute numbers and percentages. Table D.1 shows the socio-demographic details, table D.2 provides information on the relationship with cars and EVs and related aspects and also further opinions and behaviors regarding the environment and cars.

Table	Table D.1: Socio-demographic details							
Response Option	Grouped Category	ECON	(N = 100)	ENV	(N = 104)			
		Ν	%	N	%			
Age								
18 to 24 years	18 to 34 years	8	8.0	9	8.7			
25 to 34 years	18 to 34 years	33	33.0	38	36.5			
35 to 44 years	35 to 54 years	14	14.0	11	10.6			
45 to 54 years	35 to 54 years	16	16.0	21	20.2			
55 to 64 years	55 years or older	20	20.0	17	16.3			
65 years or older	55 years or older	9	9.0	8	7.7			
Gender								
Female	Female	38	38.0	36	34.6			
Male	Male	59	59.0	65	62.5			
Diverse	Diverse	2	2.0	3	2.9			
No response	No response	1	1.0	-	-			
Level of education								
Completed apprenticeship	Higher degree	17	17.0	21	20.2			
School graduation	School degree	4	4.0	5	4.8			
University degree	Higher degree	63	63.0	64	61.5			
Other degree	Other degree	3	3.0	2	1.9			
Doctorate	Highest degree	11	11.0	11	10.6			
No response	No response	2	2.0	1	1.0			
Housing situation								
Rented (apartment)	Rent	46	46.0	44	42.3			
Rented (house)	Rent	3	3.0	4	3.8			
Living with parents	Rent	2	2.0	1	1.0			
Own apartment	Ownership	4	4.0	9	8.7			
Own house	Ownership	43	43.0	44	42.3			
No information	No information	2	2.0	2	1.9			
Income								
Less than €500	Below €3,000	2	2.0	4	3.8			
€500 to under €1,000	Below €3,000	5	5.0	15	14.4			
€1,000 to under €3,000	Below €3,000	44	44.0	34	32.7			
€3,000 to under €5,000	€3,000 or above	24	24.0	23	22.1			
€5,000 and more	3,000 or above	18	18.0	16	15.4			
No response	No response	7	7.0	12	11.5			

	Telationomp, v20 concern		101		
Response Option	Grouped Category	ECON (N =	100)	ENV	(N = 104)
		N	%	Ν	%
Comparing electricity tariffs					
Yes, regularly	Compare	34	34.0	47	45.2
Yes, rarely	Not compare	41	41.0	36	34.6
No	Not compare	22	22.0	15	14.4
I don't handle the electricity bill	Not compare	2	2.0	6	5.8
No response	No response	1	1.0	-	-
Technical Interest					
Not at all	Not interested	3	3.0	1	1.0
2	Not interested	9	9.0	10	9.6
3	Moderately interested	16	16.0	12	11.5
4	Moderately interested	31	31.0	27	26.0
Very	Very interested	41	41.0	54	51.9
No response	No response	-	-	1	1.0
Concerned about climate change					
Not at all	Not concerned	2	2.0	3	2.9
2	Not concerned	4	4.0	4	3.8
3	Moderately concerned	11	11.0	5	4.8
4	Moderately concerned	28	28.0	39	37.5
Very	Very concerned	55	55.0	52	50.0
No response	No response	-	-	1	1.0
Car relationship (multiple responses possible)					
Emotional	Emotional	18	18.0	18	173
Neutral	Neutral	10	10.0	10	18.3
Functional	Functional	71	71.0	72	69.2
Rejecting	Rejecting	8	8.0	8	7.7
			0.0	-	
Car usage		22	22.0	10	15.0
Several times a day	Frequently	22	22.0	18	17.3
Once a day	Frequently	21	21.0	24	23.1
3 to under 7 times a week	Infrequently	11	11.0	22	21.2
I to under 3 times a week	Infrequently	28	28.0	21	20.2
Not at all	No non arres	1	17.0	17	10.5
No response	No response	1	1.0	4	1.9
Distance travelled					
0 km	No distances	20	20.0	22	21.2
1 km to under 50 km	Short distances	49	49.0	56	53.8
50 km to under 200 km	Long distances	28	28.0	21	20.2
No response	No response	3	3.0	5	4.8
EV Ownership					
Yes (owned/ leased/ company car)	Owner	30	30.0	31	29.8
No. but I plan to buy one	Interested	10	10.0	11	10.6
No, I don't plan to, but I don't rule it out	Interested	47	47.0	52	50.0
No, and I don't want one	Not interested	8	8.0	8	7.7
No response	No response	5	5.0	2	1.9
V2C femilierity					
V2G familiarity	Not familian	20	20.0	26	25.0
	Familiar	29	29.0	20	25.0
2	Familiar	8	8.0	7	6.7
5 A	Familiar	28	28.0	10	18.3
very well	Very familiar	20	20.0 24.0	43	41.3
No response	No response	-		2	1.9
	T T T				
V2G concerns (multiple responses possible)	D. H				10.0
Battery usage	Battery usage	17	17.0	11	10.6
Duty to connect	Duty to connect	13	13.0	10	9.6
Range anxiety	Range anxiety	12	12.0	9	8.7
Complexity	Complexity	0	6.0	3	2.9
Digitization	Digitization	∠	2.0	2	1.9
Data snaring Financial banefit is doubted	Data snaring	4	4.0	4	3.8
Financial benefit is doubted	Financial benefit is doubted	2	2.0	4	1.9
External interference/ loss of control	External interference	4	2.0	-	-

Table D.2: Car and EV relationship, V2G concerns and behavior

E. Utility functions

This section presents the utility functions formulated for the Multinomial Logit (MNL) model in detail.

This is followed by the utility functions for the Mixed Logit (MXL) model.

```
E.1. Multinomial Logit Model
V1 = ASC_V1 + bMinPlugIn_10h * MinPlugIn_10h1 + bMinPlugIn_15h * MinPlugIn_15h1 \
   + bChangeContractFlexibly * ChangeContractFlexibly1 \
  + bMinBattlevel_35p * MinBattlevel_35p1 + bMinBattlevel_50p * MinBattlevel_50p1 \
   + bVarRem_5c * VarRem_5c1 + bVarRem_10c * VarRem_10c1
   + bIncome_VarRem_5c * VarRem_5c1 * IncomeBelow3000 \
   + bIncome_FixRem * FixRem1 * IncomeBelow3000 \
   + bFixRem * FixRem1 \
   + bFemale * Female + bDiverse * Diverse \
   + bAge_18_34 * Age_18_34 + bAge_35_54 * Age_35_54 \
   + bInfrequentDrivers * InfrequentDrivers + bDoNotDrive * DoNotDrive \
   + bShortDistances * ShortDistances + bLongDistances * LongDistances \
   + bHigherDegree * HigherDegree + bHighestDegree * HighestDegree \
   + bOwner * Owner + bEV_NotInterested * EV_NotInterested \
   + bIncomeBelow3000 * IncomeBelow3000 + bNoIncomeGroup * NoIncomeGroup \
   + bSkimmedContractAttributes * SkimmedContractAttributes \
   + bCompareTariffNotAtAll * CompareTariffNotAtAll \
   + bModeratelyInterested * ModeratelyInterested + bVeryInterested * VeryInterested \
   + bFamiliarWithV2G * FamiliarWithV2G + bVeryFamiliarWithV2G * VeryFamiliarWithV2G \
   + bModeratelyConcerned * ModeratelyConcerned + bVeryConcerned * VeryConcerned \
   + bOwnership * Ownership \
   + bEmotional * Emotional \
   + bNeutral * Neutral \setminus
   + bFunctional * Functional \
   + bRejecting * Rejecting \setminus
   + bBatteryDegradation * BatteryDegradation \
   + bComplexity * Complexity \setminus
   + bDuty * Duty \
   + bRangeAnxiety * RangeAnxiety \
   + bNudge Env * Nudge Env \
   + bNudge_Familiar * Nudge_Env * FamiliarWithV2G
V2 = ASC_V2 + bMinPlugIn_10h * MinPlugIn_10h2 + bMinPlugIn_15h * MinPlugIn_15h2 \
   + bChangeContractFlexibly * ChangeContractFlexibly2 \
   + bMinBattlevel_35p * MinBattlevel_35p2 + bMinBattlevel_50p * MinBattlevel_50p2 \
   + bVarRem_5c * VarRem_5c2 + bVarRem_10c * VarRem_10c2 \
   + bIncome_VarRem_5c * VarRem_5c2 * IncomeBelow3000 \
   + bIncome_FixRem * FixRem2 * IncomeBelow3000 \
   + bFixRem * FixRem2 \
   + bFemale * Female + bDiverse * Diverse \
   + bAge_18_34 * Age_18_34 + bAge_35_54 * Age_35_54 \
   + bInfrequentDrivers * InfrequentDrivers + bDoNotDrive * DoNotDrive \
   + bShortDistances * ShortDistances + bLongDistances * LongDistances \
   + bHigherDegree * HigherDegree + bHighestDegree * HighestDegree \
   + bOwner * Owner + bEV_NotInterested * EV_NotInterested \
   + bIncomeBelow3000 * IncomeBelow3000 + bNoIncomeGroup * NoIncomeGroup \
   + bSkimmedContractAttributes * SkimmedContractAttributes \
   + bCompareTariffNotAtAll * CompareTariffNotAtAll \
   + bModeratelyInterested * ModeratelyInterested + bVeryInterested * VeryInterested \
   + bFamiliarWithV2G * FamiliarWithV2G + bVeryFamiliarWithV2G * VeryFamiliarWithV2G \
   + bModeratelyConcerned * ModeratelyConcerned + bVeryConcerned * VeryConcerned \
   + bOwnership * Ownership \
   + bEmotional * Emotional \
   + bNeutral * Neutral \setminus
   + bFunctional * Functional \
   + bRejecting * Rejecting \
   + bBatteryDegradation * BatteryDegradation \
   + bComplexity * Complexity \
```

```
+ bDuty * Duty \
   + bRangeAnxiety * RangeAnxiety \
   + bNudge_Env * Nudge_Env \
   + bNudge_Familiar * Nudge_Env * FamiliarWithV2G
V3 = ASC_no + 0
E.2. Mixed Logit Model
V1 = ASC_V1 + bMinPlugIn_10h_RND * MinPlugIn_10h1 + bMinPlugIn_15h_RND * MinPlugIn_15h1 \
   + bChangeContractFlexibly_RND * ChangeContractFlexibly1 \
   + bVarRem_5c_RND * VarRem_5c1 + bVarRem_10c_RND * VarRem_10c1
   + bFixRem_RND * FixRem1 \
   + bMinBatterylevel_35p_RND * MinBatterylevel_35p1 \
   + bMinBatterylevel_50p_RND * MinBatterylevel_50p1 \
   + bFemale * Female + bDiverse * Diverse \
   + bAge_18_34 * Age_18_34 + bAge_35_54 * Age_35_54 \
   + bOwner * Owner + bEV_NotInterested * EV_NotInterested \
   + bIncomeBelow3000 * IncomeBelow3000 + bNoIncomeGroup * NoIncomeGroup \
   + bSkimmedContractAttributes * SkimmedContractAttributes \
   + bCompareTariffNotAtAll * CompareTariffNotAtAll \
   + bModeratelyInterested * ModeratelyInterested + bVeryInterested * VeryInterested \
   + bFamiliarWithV2G * FamiliarWithV2G + bVeryFamiliarWithV2G * VeryFamiliarWithV2G \
   + bModeratelyConcerned * ModeratelyConcerned + bVeryConcerned * VeryConcerned \
   + bOwnership * Ownership \
   + bEmotional * Emotional \
   + bNeutral * Neutral \
   + bFunctional * Functional \
   + bRejecting * Rejecting \
   + bBatteryDegradation * BatteryDegradation \
   + bComplexity * Complexity \
   + bDuty * Duty \
   + bRangeAnxiety * RangeAnxiety \
   + bNudge_Env * Nudge_Env
V2 = ASC_V2 + bMinPlugIn_10h_RND * MinPlugIn_10h2 + bMinPlugIn_15h_RND * MinPlugIn_15h2 \
   + bChangeContractFlexibly_RND * ChangeContractFlexibly2 \
   + bVarRem_5c_RND * VarRem_5c2 + bVarRem_10c_RND * VarRem_10c2 \
   + bFixRem_RND * FixRem2 \setminus
   + bMinBatterylevel_50p_RND * MinBatterylevel_50p2 \
   + bMinBatterylevel_70p_RND * MinBatterylevel_70p2 \
   + bFemale * Female + bDiverse * Diverse \setminus
   + bInfrequentDrivers * InfrequentDrivers + bDoNotDrive * DoNotDrive \
   + bShortDistances * ShortDistances + bLongDistances * LongDistances \
   + bHigherDegree * HigherDegree + bHighestDegree * HighestDegree \
   + bAge_18_34 * Age_18_34 + bAge_35_54 * Age_35_54 \
   + bOwner * Owner + bEV_NotInterested * EV_NotInterested \
   + bIncomeBelow3000 * IncomeBelow3000 + bNoIncomeGroup * NoIncomeGroup \
   + bSkimmedContractAttributes * SkimmedContractAttributes \
   + bCompareTariffNotAtAll * CompareTariffNotAtAll \
   + bModeratelyInterested * ModeratelyInterested + bVeryInterested * VeryInterested \
   + bFamiliarWithV2G * FamiliarWithV2G + bVeryFamiliarWithV2G * VeryFamiliarWithV2G \
   + bModeratelyConcerned * ModeratelyConcerned + bVeryConcerned * VeryConcerned \
   + bOwnership * Ownership \
   + bEmotional * Emotional \
   + bNeutral * Neutral \setminus
   + bFunctional * Functional \
   + bRejecting * Rejecting \
   + bBatteryDegradation * BatteryDegradation \
   + bComplexity * Complexity \
   + bDuty * Duty \
   + bRangeAnxiety * RangeAnxiety \
   + bNudge_Env * Nudge_Env
```

```
V3 = ASC_no + 0
```

F. Complete list of socio-demographic and characteristic influences

This section presents the influence of all socio-demographic factors and characteristics on the willingness to participate in V2G. Therefore, also non-significant influences are shown in Table F.1.

		MNL	Model		MXL	Model	
Category	Variable	β -value	Rob. SE	Sig.	β -value	Rob. SE	Sig.
Age	18-34	0.126	0.247		0.306	0.262	
0	35-54	-0.211	0.155		-0.127	0.166	
Gender	Female	0.028	0.169		-0.082	0.182	
	Diverse	-0.556	0.406		-0.511	0.432	
Level of education	Higher Degree	0.124	0.297		0.102	0.258	
	Highest Degree	-0.122	0.320		-0.047	0.294	
Housing situation		-0.237	0.199		-0.263	0.219	
Income	Below €3,000	0.519	0.191	**	0.528	0.166	**
	No income group	-0.435	0.205	*	-0.376	0.214	
Car relationship	Emotional	0.153	0.193		0.258	0.212	
	Neutral	0.777	0.216	***	0.944	0.244	***
	Functional	0.179	0.185		0.294	0.207	
	Rejecting	1.232	0.413	**	1.856	0.382	***
Comparing electricity tariffs	Not compare	0.390	0.132	**	0.425	0.143	**
Technical Interest	Moderately interested	0.041	0.227		-0.108	0.236	
	Very interested	0.217	0.276		0.104	0.287	
Concerned about climate change	Moderately concerned	0.087	0.254		0.065	0.258	
	Very concerned	0.095	0.253		0.107	0.252	
Car usage	Infrequently	-0.184	0.154		0.213	0.135	
Distance travelled	Short distances	-0.297	0.225		-0.095	0.178	
	Long distances	-0.264	0.238		-0.108	0.208	
EV ownership	Owner	-0.700	0.154	***	-0.689	0.167	***
	Not interested in EV	-1.535	0.221	***	-1.504	0.237	***
V2G familiarity	Familiar	0.174	0.181		0.240	0.168	
	Very familiar	0.470	0.203	*	0.473	0.211	*
Concerns	Battery degradation	-0.491	0.197	*	-0.426	0.224	
	Complexity	-0.690	0.273	*	-1.096	0.346	**
	Range anxiety	-0.685	0.285	*	-0.853	0.328	**
	Duty	0.373	0.328		0.596	0.365	
Attribute description	Skimmed	-0.024	0.130		-0.025	0.143	

Table F.1: Socio-demographic and characteristic influences

Significance levels: *** p<0.001, ** p<0.01, * p<0.05

G. MNL results for both income groups

To isolate group-specific differences, separate MNL models were estimated for each income group. The results are summarized in Table G.1.

		Income	< €3,000 /m	nonth net	Income	≥ €3,000 /m	nonth net
Attribute	Level	β -value	Rob. SE	Sig.	β -value	Rob. SE	Sig.
Fixed remuneration	0-50 cents/day	0.023	0.002	***	0.016	0.002	***
Variable remuneration	5 cents/kWh	-0.394	0.081	***	-0.822	0.094	***
	10 cents/kWh	-0.038	0.089		-0.104	0.102	
Contract flexibility	Monthly	-0.355	0.068	***	-0.434	0.082	***
Minimum plug-in duration	10 hours	-0.199	0.080	*	-0.175	0.094	
Minimum battery level	35%	-0.111	0.087		-0.362	0.102	***
	50%	-0.216	0.084	*	-0.279	0.096	***
Model Fit		Value			Value		
\mathbb{R}^2		0.335			0.233		
AIC		2361			2128		
BIC		2574			2332		

Table G.1: Results of the MNL model: Willingness to accept

Significance levels: *** p<0.001, ** p<0.01, * p<0.05

H. Attributes from literature

Table H.1 summarizes the properties and key assumptions of the four main relevant choice studies in the context of V2G.

	Country	Remuneration	Min. range/ battery level	Plug-in duration	No. of Cycles	Contract Duration	Order of the Attributes Importance
Parsons et al. (2014)	US	\$500/year \$1,000/year \$2,000/year \$3,000/year \$4,000/year \$5,000/year	25 miles 75 miles 125 miles 175 miles	5 hours/day 10 hours/day 15 hours/day 20 hours/day			 Remuneration Min. range/ battery level Plug-in duration
Geske and Schumann (2018)	DE	15/month 30/month 45/month 60/month	10 km 20 km 30 km 40 km 50 km	0 hours/day 5 hours/day 7 hours/day 10 hours/day 14 hours/day			 Min. range/ battery level Plug-in duration Remuneration
Kubli et al. (2018)	СН	$\begin{array}{l} Cost \ reduction \ of: \\ CHF60/month \\ CHF40/month \\ CHF20/month \end{array}$	40% 60% 80% 100%	, .	0/day 1/day 3/day Unlimited	0 months 12 months 24 months 48 months	 Remuneration Min. range/ battery level Discharging cycles Contract duration
Huang et al. (2021)	NL	$\begin{array}{l} { { $	10% 30% 50%	5 hours/day 10 hours/day 15 hours/day	1/session 4/session 7/session	6 months 12 months 24 months	 Min. range/ battery level Plug-in duration Fixed remuneration Discharging cycles

\mathbf{T}
